A randomised, parallel-group, double-blind, placebo-controlled phase 3 study to Determine the effectiveness of the type I interferon receptor antibody, Anifrolumab, In SYstemic sclerosis: DAISY study design and rationale

D. Khanna¹, C.P. Denton², S. Assassi³, M. Kuwana⁴, Y. Allanore⁵, R.T. Domsic⁶, C. Kleoudis⁷, J. Xu⁷, E. Csomor⁸, C. Seo⁹, M. Albulescu⁸, R. Tummala⁷, H. Al-Mossawi⁸, R.N. Kalyani⁷, F. Del Galdo¹⁰

 ¹University of Michigan Scleroderma Program, Ann Arbor, MI, USA; ²Centre for Rheumatology, Division of Medicine, University College London, UK; ³McGovern Medical School, Division of Rheumatology, University of Texas Health Science Center at Houston, TX, USA; ⁴Department of Allergy and Rheumatology, Nippon Medical School, Tokyo, Japan; ⁵Rheumatology Department, Cochin Hospital, Université Paris Cité, INSERM U1016; Paris, France; ⁶University of Pittsburgh School of Medicine, Pittsburgh, PA, USA;
 ⁷BioPharmaceuticals R&D, AstraZeneca, Gaithersburg, MD, USA; ⁸BioPharmaceuticals R&D, AstraZeneca, Cambridge, UK; ⁹BioPharmaceuticals Medical Evidence, AstraZeneca, Gaithersburg, MD, USA; ¹⁰Department of Rheumatology, Leeds Institute of Rheumatic and Musculoskeletal Medicine, University of Leeds, UK.

Abstract Objective

The type I interferon pathway is a promising target for treatment of patients with systemic sclerosis (SSc). Here, we describe the design of a multinational, randomised phase 3 study to Determine the effectiveness of the type I interferon receptor antibody, Anifrolumab, In SYstemic sclerosis (DAISY).

Methods

DAISY includes a 52-week double-blind, placebo-controlled treatment period, a 52-week open-label active treatment period, and a 12-week safety follow-up period. The patient population includes a planned 306 adults with limited or diffuse cutaneous active SSc who satisfied American College of Rheumatology/European Alliance of Associations for Rheumatology 2013 SSc criteria. Use of standard immunosuppressants, including mycophenolate mofetil, at a stable dose prior to randomisation is permitted in addition to weekly subcutaneous anifrolumab or placebo. Efficacy will be assessed at Week 52 via Revised-Composite Response Index in SSc (CRISS)-25 response (primary endpoint).
 Lung function and skin thickness will be assessed via change from baseline in forced vital capacity in patients with SSc-associated interstitial lung disease and modified Rodnan Skin Score, respectively (key secondary endpoints).

Conclusion

The DAISY trial will evaluate the efficacy and safety of anifrolumab as a first-in-class treatment option for patients with both limited and diffuse cutaneous SSc and will provide insight into the contributions of type I interferon to SSc pathogenesis. Revised-CRISS-25 can account for improvement and worsening in a broad set of validated clinical measures beyond lung function and skin thickness, including clinician- and patient-reported outcomes, capturing the heterogeneity of SSc.

Key words

autoimmune disease, systemic sclerosis, anifrolumab, type I interferon

Dinesh Khanna, MD, MSc Christopher P. Denton, FRCP Shervin Assassi, MD, MS Masataka Kuwana, MD, PhD Yannick Allanore, MD, PhD Robyn T. Domsic, MD, MPH Christi Kleoudis, MPH John Xu, PhD Eszter Csomor, PhD Caroline Seo Marius Albulescu, MD Raj Tummala, MD, MBA* Hussein Al-Mossawi, MD, PhD* Rubana N. Kalyani Francesco Del Galdo, MD, PhD *Affiliation at the time of study design.

Please address correspondence to: Dinesh Khanna University of Michigan Scleroderma Program, 1500 E Medical Center Dr, Ann Arbor, MI 48109, USA. E-mail: khannad@med.umich.edu

and to: Francesco Del Galdo Leeds Institute of Rheumatic and Musculoskeletal Medicine, NIHR Biomedical Research Centre, University of Leeds, 2nd Floor, Chapel Allerton Hospital, LS7 4SA, Leeds, UK. E-mail: f.delgaldo@leeds.ac.uk Received on April 24, 2024; accepted in revised form on June 10, 2024. © Copyright CLINICAL AND EXPERIMENTAL RHEUMATOLOGY 2024.

Competing interests: page 1642.

Plain language summary

Systemic sclerosis is a chronic autoimmune disease that leads to inflammation and scarring of the skin and internal organs, especially the lungs. Systemic sclerosis and lupus are both associated with increased interferon signalling, which is usually triggered by viral infections, but is related to damaging inflammation in these diseases. Anifrolumab, a drug that blocks interferon signalling, is already used to treat patients with lupus (also known as SLE), so it could potentially be used to treat patients with systemic sclerosis. This publication details the DAISY study design and explains why it is needed.

This study will follow 2 groups of 153 patients with systemic sclerosis over 2 years. During the first year, in addition to any standard immunosuppressant therapy, the groups will receive weekly injections of either anifrolumab or "dummy drug" (placebo). In the second year, all patients will receive anifrolumab with their standard immunosuppressant therapy. Multiple factors will be considered to evaluate the efficacy of anifrolumab treatment, including clinical measurements of skin thickness and lung function, and questionnaires completed by clinicians and patients to report on patient health and their everyday function during treatment.

The DAISY study will investigate the efficacy and safety of anifrolumab treatment in a diverse group of patients with systemic sclerosis who currently have limited options for effective treatment. The study will evaluate the impact of anifrolumab treatment on multiple aspects of the disease, and how patients feel about their overall health-related quality of life.

Introduction

Systemic sclerosis (SSc) is a complex, heterogeneous autoimmune disease in which vasculopathy, inflammation, and fibrosis contribute to damage in multiple organs (1). Dysregulated interferon signalling is a cardinal feature of autoimmune diseases including systemic lupus erythematosus (SLE), Sjögren's syndrome, and SSc (2). There is evidence that type I interferon dysregulation may play an important role in SSc

pathogenesis (3). Excess type I interferon signalling has been consistently identified in the blood and tissues in the majority of patients with SSc(4, 5), and the degree of type I interferon pathway activity has been associated with the presence of SSc autoantibodies (6) and the severity of organ involvement (3, 7). Immunoablation and stem cell rescue in patients with SSc was observed to normalise the levels of two interferon gene expression modules; this change in transcriptional levels was significantly correlated with improvement in forced vital capacity (FVC) (8), an important clinical measure of lung function (9).

Particularly strong support for a pathogenic role of type I interferon in SSc came from a randomised, placebo-controlled trial of patients with early diffuse cutaneous SSc, in which treatment with recombinant type I interferon- α led to SSc disease progression rather than the disease improvement that had been anticipated (10). Conversely, inhibition of the type I interferon pathway has demonstrated efficacy in the treatment of SLE, another disease in which the interferon signature has been associated with disease activity (11).

Anifrolumab is a fully-human, IgG1 α monoclonal antibody that targets the type I interferon receptor α subunit 1 (IFNAR1) and blocks downstream type I interferon signalling, which suppresses the interferon gene signature (IFNGS) (12-14). Following phase 3 trials, anifrolumab was approved for use in patients with moderate to severe SLE in multiple countries (14-19); the use of anifrolumab in patients with SLE is supported by 4 years of safety data (20).

The safety/tolerability, pharmacodynamics, and pharmacokinetics of anifrolumab were previously investigated in a phase 1 trial of patients with SSc (21). This study revealed an adequate safety and tolerability profile of anifrolumab in patients with SSc, whereby most treatment-emergent adverse events were mild or moderate, and anifrolumab treatment decreased type I IFNGS expression in whole blood and skin (21). IFNAR1 blockade was also associated with suppression of T-cell activation markers and collagen accumulation, suggesting the potential benefit of ani-



Fig. 1. Study design.

Figure originally published in Khanna D et al.: J Scleroderma Relat Disord 2024; 9(Supplement): 61-395 (39).

The DAISY study will include adults with limited or diffuse cutaneous active SSc who were ≤ 6 years from their first non-Raynaud's manifestation of the disease. Patients are permitted to take standard immunosuppressant therapy (hydroxychloroquine, MMF, MPA/MPS, methotrexate, azathioprine, tacrolimus, or oral glucocorticoids) in addition to randomized treatment. A group of 15 patients with SSc in the United States and the United Kingdom were consulted on study design, visit schedule, self-administration dosing, and visit procedures/activities.

^aThe last dose of study intervention will be administered at Week 51 in the double-blind treatment period and at Week 103 in the open-label treatment period; Week 52 is a shared visit between the double-blind treatment period (procedures) and the open-label treatment period (study drug administration); ^bRevised version of the 2-step ACR-CRISS algorithm.

ACR: American College of Rheumatology; CRISS: Composite Response Index in Systemic Sclerosis; DAISY: Determine the effectiveness of the type I interferon receptor antibody, Anifrolumab, In SYstemic sclerosis; MMF: mycophenolate mofetil; MPA/MPS: mycophenolic acid or mycophenolate sodium; QW: once weekly; R: randomisation; SC: subcutaneous; SSc: systemic sclerosis.

frolumab for targeting tissue fibrosis, a hallmark of SSc pathology (22).

Together, the strong evidence for the role of type I interferon in SSc pathogenesis and the safety and efficacy results from the phase 1 trial in patients with SSc support further investigation of anifrolumab as a treatment for SSc (3, 21, 22). Thus, here we report the study design of a multinational, randomised, placebocontrolled, double-blind phase 3 study (DAISY; NCT05925803). DAISY aims to evaluate the efficacy of subcutaneous (SC) anifrolumab in adult patients with SSc who may be taking one or a combination of protocol-specified standard therapies and to establish a safety profile of anifrolumab treatment in SSc.

Methods

Study design

DAISY is a multinational, randomised (1:1), parallel-group, double-blind, placebo-controlled, phase 3 study that is planned to include 306 patients. The study includes 17 study visits over 4 periods with a total study duration of ~122 weeks: a <6-week screening period prior to randomisation; a 52-week double-blind, placebo-controlled treatment period; a 52-week open-label active treatment period; and a 12-week safety follow-up period (Fig. 1). The primary and key secondary endpoints will be evaluated at Week 52. The open-label treatment period will allow for the observation of treatment effects

beyond 1 year. Patient input contributed to the design of the study.

Ethics

The study is designed to be conducted in accordance with consensus ethical principles derived from international guidelines, including the Declaration of Helsinki as amended in October 2013. The study protocol, any amendments, Investigator's Brochure, and informed consent forms will be reviewed and approved by an Institutional Review Board/Ethics Committee for each study site, and patients will provide informed consent prior to participation.

Patient selection

DAISY will include adults 18-70 years of age (inclusive) with SSc (disease duration within 6 years from first non-Raynaud's phenomenon manifestation) according to 2013 American College of Rheumatology/European Alliance of Associations for Rheumatology (ACR/ EULAR) classification criteria for SSc (23), and either limited or diffuse cutaneous subset as defined by LeRoy and Medsger (24). Use of standard immunosuppressants, including hydroxychloroquine, methotrexate, azathioprine, mycophenolate mofetil (MMF), mycophenolic acid or mycophenolate sodium (MPA/MPS), tacrolimus, or oral glucocorticoids, will be permitted (not required) at a stable dose prior to randomisation; the dose should remain stable throughout the study. Key inclusion and exclusion criteria are presented in Table I.

Treatment

During the 52-week double-blind period, study participants will receive onceweekly treatment with anifrolumab, administered subcutaneously via an accessorised prefilled syringe (0.8 mL fill volume), or matching placebo. During the 52-week open-label period, all participants will receive once-weekly treatment with anifrolumab, administered subcutaneously. Study intervention may be administered by study staff, patient, or caregiver, in the clinic or, for weekly doses in between clinic visits, at home.

Efficacy assessments

Study objectives and endpoints are outlined in Table II and endpoint definitions are included in Table III.

The primary efficacy endpoint will be Revised-Composite Response Index in SSc (CRISS)-25 (Revised-CRISS-25) response (yes/no) at Week 52 (25). Revised-CRISS-25 uses a 2-step algorithm, with equal weighting given to each of the five validated core measures (modified Rodnan skin score [mRSS], percent predicted forced vital capacity [ppFVC], Health Assessment Questionnaire-Disability Index [HAQ-DI], and patient and clinician global assessments [PtGA and CGA]) (25, 26) and assesses percent improvement and worsening

Table I. Key inclusion and exclusion criteria.

Inclusion criteria

Demographic characteristics

Adult patients from 18 to 70 years of age inclusive

Disease characteristics

SSc according to 2013 ACR/EULAR classification criteria

Limited or diffuse cutaneous subsets according to LeRoy and Medgser

SSc disease duration within 6 years from first non-Raynaud's phenomenon manifestation at the time of signing the ICF

Either HAQ-DI score ≥0.25 points or PtGA score ≥3 points

mRSS >10 with disease duration <18 months and/or lung involvement (ILD) as defined by the protocol, OR

If disease duration \geq 18 months and patient does not have ILD, mRSS must be \geq 15 with active disease at screening with at least one of the following:

1) C-reactive protein ≥ 0.6 mg/dL that is unrelated to other conditions or erythrocyte sedimentation rate ≥ 28 mm/hr or platelet count $\ge 330 \times 10^9$ /L;

2) new skin involvement or skin progression by mRSS \geq 3 units compared with recent assessments performed within the previous 6 months; or

3) presence of at least one tendon friction rub documented in medical records within the previous 3 months of screening or at screening

Background therapies

Stable background therapies can be used including hydroxychloroquine ($\leq 400 \text{ mg/day}$), methotrexate ($\leq 25 \text{ mg/week}$), azathioprine ($\leq 200 \text{ mg/day}$), MMF ($\leq 3 \text{ g/day}$), MPA/MPS ($\leq 2.16 \text{ g/day}$), oral glucocorticoids ($\leq 10 \text{ mg/day}$ prednisone or equivalent), or tacrolimus ($\leq 0.2 \text{ mg/kg/day}$)*

Sex and contraceptive/barrier requirements

Women of childbearing potential with a negative urine pregnancy test

Other criteria

Uninvolved or mildly thickened skin at injection sites Informed consent

Exclusion criteria

Anticentromere antibody seropositivity on central laboratory

History of SSc renal crisis within past 12 months or eGFR<45 mL/min/1.73 m²

Overlap syndromes, systemic lupus erythematosus with anti-dsDNA antibody seropositivity or anti-citrullinated protein antibodies-positive rheumatoid arthritis, or SSc mimics (*e.g.* scleromyxedema, eosinophilic fasciitis)

History of, or current, other inflammatory diseases (*e.g.* inflammatory bowel disease, skin disease) that, in the opinion of the investigator, could interfere with efficacy and safety assessments or require immunomodulatory therapy

Evidence of moderately severe concurrent nervous system, renal, endocrine, hepatic, or gastrointestinal disease (*e.g.* clinical signs of malabsorption or needing parenteral nutrition) not related to SSc, as determined by the investigator

Pulmonary disease with FVC \leq 50% predicted, DL_{co} haemoglobin corrected \leq 45% predicted, or airway obstruction (pre-bronchodilator FEV1/FVC <0.7) at screening

Evidence of other severe pulmonary disease as determined by the investigator (*e.g.* asthma requiring biologic therapy, chronic obstructive pulmonary disease requiring long-term, at-home oxygen)

Cardiovascular disease with significant arrhythmia, congestive heart failure (NYHA Class II-IV), unstable angina, uncontrolled hypertension, cor pulmonale, symptomatic pericardial effusion, or cardiac abnormality such as left ventricular failure with ejection fraction <50%

Significant pulmonary hypertension

Haematopoietic stem cell transplantation or solid organ/limb transplantation

Any severe case of herpes zoster infection as defined by the protocol

Known malignancy or a history of malignancy within 5 years, with exception of excised/cured local basal or squamous cell carcinoma of the skin or cervical cancer in situ

Major surgery within 8 weeks prior to and/or during study enrolment

Known active current or history of recurrent infections

Any condition that, in the opinion of the investigator or AstraZeneca, would interfere with the efficacy or safety evaluation of the study intervention or put participant at safety risk

*If their doses remained stable for \geq 3 months prior to randomization, patients are permitted to have used MMF or MPA/MPS, azathioprine, methotrexate, or tacrolimus in combination with hydroxychloroquine and/or low-dose oral glucocorticoids (\leq 10 mg/day prednisone or equivalent). Patients receiving low-dose glucocorticoids must have maintained a stable dose for \geq 2 weeks prior to randomisation. Combinations of MMF, MPA/MPS, azathioprine, methotrexate, or tacrolimus with each other or with other conventional immunosuppressants are not permitted.

ACR: American College of Rheumatology; anti-dsDNA: anti-double-stranded DNA; DL_{CO}: diffusing capacity of the lung for carbon monoxide; eGFR: estimated glomerular filtration rate; EULAR: European Alliance of Associations for Rheumatology; FVC: forced vital capacity; HAQ-DI: Health Assessment Questionnaire-Disability Index; ICF: informed consent form; ILD: interstitial lung disease; MMF: mycophenolate mofetil; MPA/MPS: mycophenolic acid or mycophenolate sodium; mRSS: modified Rodnan Skin Score; NYHA: New York Heart Association; PtGA: Patient's Global Assessment.

in these measures in patients with SSc (Fig. 1). Significant SSc-related events are adjudicated according to the following definitions in Step 1: Renal) new onset scleroderma renal crisis; interstitial lung disease [ILD]) new decline in ppFVC $\geq 15\%$ (relative to baseline

assessment) in new or established ILD and new ppFVC <80% predicted; heart) new onset left ventricular failure (left ventricular ejection fraction \leq 45%) requiring treatment; cardiopulmonary) new onset pulmonary arterial hypertension on right heart catheterisation requiring treatment; gastrointestinal) new significant dysmotility requiring enteral (such as nasogastric, jejunostomy, or gastric tube) or parenteral nutrition except for oral supplements; or digital ischaemia) new gangrene, amputation, or unscheduled hospitalisation due to

Table II. Study objectives and endpoints.

Objective	Endpoints
Primary	
To demonstrate the superiority of anifrolumab compared with placebo with or without standard therapy on measures of signs, symptoms, and impacts associated with SSc	 Revised-CRISS-25 response at week 52 Where a responder is defined as a patient who meets all of the following criteria: Improvement in at least two components: ≥5% increase for ppFVC and/or ≥25% decrease for mRSS, HAQ-DI, PtGA, CGA Worsening in no more than one component: ≥5% decrease for ppFVC and/or ≥25% increase for mRSS, HAQ-DI, PtGA, CGA No significant SSc-related event ^a Otherwise, a patient is a non-responder
Key secondary	
To demonstrate the superiority of anifrolumab to placebo with or without standard therapy on lung function in patients with SSc-associated ILD	Change from baseline in FVC (mL) at week 52
To demonstrate the superiority of anifrolumab to placebo with or without standard therapy on skin thickness in patients with SSc	Change from baseline in mRSS at week 52
Secondary	
To assess the effect of anifrolumab compared with placebo with or without standard therapy on each component of the measures of signs, symptoms, and impacts associated with SSc included in the Revised-CRISS-25 response	Improvement in each component (evaluated separately) of the Revised-CRISS-25 at week 52, where a responder is defined as follows: • ppFVC ≥5% increase • mRSS ≥25% decrease • HAQ-DI ≥25% decrease • PtGA ≥25% decrease • CGA ≥25% decrease • CGA ≥25% decrease Otherwise, a participant is a non-responder for the component
To assess the effect of anifrolumab compared with placebo with or without standard therapy on ILD progression as quantified by CT in patients with SSc	Change from baseline at week 52 in CT measures of interstitial lung disease (Quantitative ILD, Quantitative light-induced fluorescence)
To assess the effect of anifrolumab compared with placebo with or without standard therapy on skin-specific HRQoL assessment in patients with SSc	Change from baseline in SSPRO at week 52
To assess the effect of anifrolumab compared with placebo with or without standard therapy on lung function in patients with SSc	Change from baseline in ppFVC at week 52 Change from baseline in FVC (mL) at week 52
To assess the effect of anifrolumab compared with placebo with or without standard therapy on lung function in patients with SSc-associated ILD	Change from baseline in ppFVC at week 52
To evaluate the pharmacokinetics, pharmacodynamics, and immunogenicity of SC anifrolumab	Anifrolumab concentration and pharmacokinetic parameters, anti-drug antibodies, and type I interferon 21-gene signature from blood Anti-drug antibody presence and titre during treatment and follow-up Relationships between development of anti-drug antibodies and efficacy, safety, or pharmacokinetic outcome measures

^a Patient is considered to have a significant SSc-related event, irrespective of improvement in other core components, if they develop any one of the following during the study as defined in Step 1: (1) Renal: new scleroderma renal crisis; (2) ILD: new decline in ppFVC $\geq 15\%$ (relative to baseline assessment) in new or established ILD and new ppFVC below 80% predicted; (3) Heart: new onset of left ventricular failure (defined as LVEF $\leq 45\%$) requiring treatment; (4) Cardiopulmonary: new onset of PAH on right heart catheterisation requiring treatment; (5) Gastrointestinal: new significant dysmotility requiring enteral or parenteral nutrition except for oral supplements; (6) Digital ischaemia: new gangrene, amputation or unscheduled hospitalisation due to worsening of digital ischaemia (*e.g.* due to infection, increasing number of digital ulcers, worsening of established digital ulcers, or surgery), not including scheduled prostacyclin infusions.

CGA: clinician's global assessment; FVC: forced vital capacity; HAQ-DI: Health Assessment Questionnaire-Disability Index; HRQoL: health-related quality of life; ILD: interstitial lung disease; mRSS: modified Rodnan skin score; ppFVC: percent predicted FVC; PtGA: Patient's Global Assessment; Revised-CRISS-25: Revised-Composite Response Index in SSc Score; SSc: systemic sclerosis; SSPRO: Scleroderma Skin Patient Reported Outcome.

worsening of digital ischaemia (*e.g.* due to infection, increasing number of digital ulcers, worsening of established digital ulcers, or surgery), not including scheduled prostacyclin infusions. Irrespective of improvement in other core components, patients that develop Step

1 events during the study are classified as non-responders. In patients with no new or worsening major organ involvement, a Revised-CRISS-25 response is determined in Step 2: response is achieved by showing improvement in ≥ 2 components and worsening in ≤ 1 component of the measure. Improvement and worsening thresholds were defined according to published minimal clinically important differences for each measure: a $\geq 25\%$ change in mRSS, HAQ-DI, CGA, or PtGA, or a $\geq 5\%$ change in ppFVC (25).

Table III. Definitions of assessments used in the DAISY study.

Assessment	Abbreviation	Definition
Revised-Composite Response Index in SSc Score*	Revised-CRISS-25	A composite endpoint where a responder is defined as a patient who shows improvement in at least 2 components (\geq 5% increase for ppFVC and/or \geq 25% decrease for mRSS, HAQ-DI, PtGA, CGA) and worsening in no more than one component (\geq 5% decrease for ppFVC and/or \geq 25% increase for mRSS, HAQ-DI, PtGA, CGA) with no significant SSc-related event over the duration evaluated; otherwise, a patient is a non-responder
Forced Vital Capacity†	FVC	Assessed with a spirometer according to ATS/ERS 2019 guidelines (40)
Modified Rodnan Skin Score†	mRSS	Patient's skin thickness is rated by clinical palpation using a 0–3 scale (0=normal skin; 1=mild thickness; 2=moderate thickness; 3=severe thickness with inability to pinch the skin into a fold) for each of 17 surface anatomic areas of the body: face, anterior chest, abdomen, and fingers, forearms, upper arms, thighs, lower legs, dorsum of hands and feet with right and left considered separately. Individual values are added and the sum is defined as the total skin score (41)
Health Assessment Questionnaire–Disability Index	HAQ-DI	20-item instrument with which a patient scores their difficulties in perform- ing activities of daily living over the last week in each of 8 sections (Dress- ing, Arising, Eating, Walking, Hygiene, Reach, Grip, and Activities) from 0 ("Without Any Difficulty") to 3 ("Unable to Do"). The overall score is determined by summing the highest item score in each section and dividing by 8. The Standard Disability Index will be used to adjust the score for each category based on the use of an aid, device, or assistance (42)
Patient's Global Assessment	PtGA	11-point numerical rating scale used for patient to report their overall health in the last week from 0 ("Excellent") to 10 ("Extremely Poor")
Clinician's Global Assessment	CGA	Physician assesses patient's overall health in the last week from 0 ("Excel- lent") to 10 ("Extremely Poor"), with an option to choose "Not Known" if the patient is new to the clinic
Scleroderma Skin Patient Reported Outcome	SSPRO	18-item instrument with 4 domains (Physical Effects, Physical Limitations, Emotional Effects, and Social Effects). Patient reports skin-related HRQoL over the last 4 weeks using a 7-point numerical rating scale from 0 ("Not at All") to 6 ("Very Much"), with lower score indicating higher skin-related HRQoL (43)

*The primary objective, to demonstrate the superiority of anifrolumab to placebo with or without standard therapy in SSc, was evaluated using Revised CRISS-25 response at Week 52. Improvements in each component of Revised-CRISS-25 at Week 52 were evaluated separately as secondary objectives; †One key secondary objective, to demonstrate the superiority of anifrolumab to placebo with or without standard therapy on lung function in patients with SSc-associated ILD, was evaluated using change from baseline in FVC (mL) at Week 52. The other key secondary objective, to demonstrate the superiority of anifrolumab to placebo with or without standard therapy on skin thickness in patients with SSc, was evaluated using change from baseline in mRSS at Week 52.

ATS/ERS: American Thoracic Society/European Respiratory Society; HRQoL: health related quality of life; ILD: interstitial lung disease; ppFVC: percent predicted forced vital capacity; SSc: systemic sclerosis.

Key secondary endpoints include change from baseline to Week 52 in 1) FVC (mL) in patients with SSc-associated ILD and 2) in mRSS in the full analysis dataset. Other secondary endpoints are listed in Table II.

Safety assessments

Safety and tolerability will be evaluated in terms of adverse events (AEs; number and percentage of patients reporting that event, and number of events where appropriate, for each treatment group by system organ class and preferred term), AEs of special interest (serious non-opportunistic infection, opportunistic infection, herpes zoster, tuberculosis [including latent tuberculosis], malignancy, injection site reaction, and major adverse cardiovascular events), vital signs, clinical laboratory tests, 12-lead electrocardiograms, physical examination, Columbia Suicide Severity Rating Scale (C-SSRS), and Patient Health Questionnaire Depression Scale (PHQ-8). Patients will be assessed for coronavirus disease-2019 at each clinic visit.

Pharmacodynamics

and pharmacokinetics Whole blood will be collected at screening and pre-dose for pharmacodynamic, pharmacokinetic, and/or

immunogenicity tests throughout the study. The suppression of the type I interferon 21-gene signature in peripheral blood as a percent of baseline will be assessed as a secondary endpoint to follow the biologic effect of anifrolumab on IFNAR1 throughout the study (27). A dichotomous 4-gene IFNGS test in peripheral blood will be used to measure the mRNA overexpression of type I interferon-inducible genes at baseline to evaluate if the participants had high/low IFNGS at study entry (12). This information will be used as part of an exploratory analysis to understand if interferon activity at baseline would impact clinical outcomes. In addition, whole blood, serum, plasma, and optional skin biopsy samples will be collected for biomarker measurement to explore the wider mechanism of action of anifrolumab.

Statistical methods

Approximately 460 patients will be screened to achieve 306 eligible patients for randomisation. A sample size of 153 participants per arm yields 90% power to detect a treatment difference of 18% in the proportions of patients who achieve revised-CRISS-25 response at week 52 between anifrolumab and placebo groups (two-sided α =0.05). The assumed proportion of responders (participants achieving Revised-CRISS-25 at Week 52) is 30% in the placebo group (25).

Efficacy analyses will be performed using the "full analysis dataset" of all patients who were randomised to treatment and received ≥ 1 dose of study intervention, or in the patients in the full analysis dataset with SSc-associated ILD. To control for study-wise Type I error, the primary endpoint will be tested at a 2-sided α =0.05 and, if statistically significant, then key secondary endpoints will be tested using a hierarchical sequential testing strategy in pre-defined order.

The proportions of patients achieving Revised-CRISS-25 response at Week 52 with anifrolumab versus placebo (the primary endpoint) will be compared using a Cochran-Mantel-Haenszel (CMH) approach, controlling for randomisation stratification factors (ILD [yes/no] at Week 0/Day 1; MMF or MPA/MPS use [yes/no] at Week 0/ Day 1; and disease duration from first non-Raynaud's symptom attributable to SSc [$</\geq 18$ months] at Week 0/Day 1). Patients were considered non-responders if they received restricted medication, discontinued investigational product, or died prior to Week 52.

To compare treatment groups and assess key secondary endpoints for change from baseline to Week 52 in mRSS, models for repeated measures will include baseline value, treatment group, visit, treatment x visit interaction, and randomisation stratification factors. Models for change from baseline in FVC (mL) will include only randomisation stratification factors MMF or MPA/MPS use at Week 0/ Day 1 and disease duration at Week 0/ Day 1. Multiple imputation, penalising based on poorest responses observed on placebo to reflect treatment failure, will be used for patient who received restricted medication, discontinued investigational product, or died prior to Week 52.

A "safety dataset" consisting of all patients who received ≥ 1 dose of study intervention will be used to assess safety and tolerability. Patients' results will be reported according to the treatment actually received.

Discussion

There is strong evidence for a role of type I interferon in the pathogenesis of SLE, SSc, and related connective tissue diseases (3, 7, 8, 10). Multiple lines of evidence support that type I interferon receptor inhibition with anifrolumab has the potential to be a first-in-class treatment option for patients with SSc (21, 22). Here, we describe the design of a multinational phase 3 study to evaluate the safety and efficacy and the long-term treatment profile of anifrolumab in adult patients with diffuse or limited cutaneous SSc.

The study partnered with 15 patients from the United States and United Kingdom living with SSc to capture the patient voice in the study design. Overall, patients perceived the study to be well-designed, and contributed ideas to reduce the burden of the trial and to make it fit-for-purpose for the SSc patient population. Discussions with patients also allowed us to better understand how to capture diaries and PRO instruments and how to mitigate SSc-related issues with using touchscreen digital devices. Patients will be completing several PRO instruments throughout the trial (including the HAQ-DI and PtGA that are components of the primary endpoint); an optional questionnaire will be made available at the beginning, middle, and end of the study, providing further opportunities to capture the patient voice throughout the drug development process and encourage patient feedback regarding their clinical study experience. Furthermore, including the open-label treatment period would provide longer-term safety data for patients with SSc and give patients randomised to placebo during the double-blind period the opportunity to receive active treatment.

A unique feature in the study design is the selection of Revised-CRISS-25 as the primary endpoint in this study, rather than skin or lung assessments alone, which do not adequately capture multiorgan disease activity (28). Use of a composite response measure can provide insights into changes in multiple organ systems, not just skin or other organ involvement (28), and therefore provide insights into overall patient health for a population of patients with a complex, heterogenous systemic disease. **Revised-CRISS** was recently developed and is being utilized in current phase 2 trials in patients with SSc (belimumab, rituximab, and MMF: NCT03844061; tulisokibart: NCT05270668; BI 685509: NCT05559580) and upcoming phase 3 studies. Revised-CRISS assesses new or meaningful worsening of internal organ involvement and measures the signs, symptoms, and impacts associated with SSc from the patient's and clinician's perspectives by incorporating multiple PROs, clinician-reported outcomes, and surrogate assessments of disease activity (25). Revised-CRISS is derived from ACR-CRISS, with revisions to ensure equal weighting of the core set variables and to limit the floor/ceiling effects seen with that measure (25, 26). Selection of the Revised-CRISS-25 threshold (which translates to improvement of 25% in mRSS, HAQ-DI, PGA, and CGA and improvement of 5% in ppFVC) was based on published minimal clinically important differences in patients with different rheumatic diseases, including SSc (25). The use of the Revised-CRISS-25 composite response enables inclusion of a broad population of patients with SSc, which is a strength of this study.

Whereas randomised SSc trials intending to treat overall SSc activity and skin fibrosis tend to include only patients with diffuse cutaneous SSc (28), this

study includes patients with limited or diffuse cutaneous SSc, which reflects the real-world target population of the therapy. Given that type I interferon might contribute to SSc pathogenesis across multiple different organs (29, 30), it is important to include a broad population of patients in this trial of anifrolumab. Type I interferon, measured as an elevated IFNGS, has been identified at the earliest phases of SSc, and in patients with both limited and diffuse cutaneous SSc (31). Interferonregulated gene expression has been implicated as a pathogenic pathway in patients with SSc-related ILD (29), and progression of ILD can be similar in patients from both cutaneous subsets (32). In this trial, we will quantify the degree of total lung involvement and fibrosis at baseline and at the end of the trial. Importantly, few treatment options are available for patients with limited cutaneous SSc; as such, new clinical trials are needed in this patient population (33).

Another important feature of the DAI-SY study design is that patients are permitted to receive background treatment with standard immunosuppressants alongside the trial intervention. MMF is the most common first-line standard therapy for patients with diffuse SSc (28, 34, 35); however, MMF has the potential to confound treatment responses. The option for background immunosuppressive treatment may, for example, reduce the frequency of Step 1 events in the determination of revised/ACR-CRISS response. The randomised, parallel-group SLS-II study evaluated the effects of MMF treatment for 24 months on lung function, compared with oral cyclophosphamide treatment for 12 months, in patients with SSc and symptomatic ILD (36). Though both treatment groups eventually had a similar overall maximum response, ppFVC increased earlier in patients receiving MMF (at 6-9 months) than with cyclophosphamide (34, 36). In the RESOLVE-1 trial, patients were permitted to continue established treatment with stable doses of concomitant immunosuppressives (defined as no dose increases within 8 weeks prior to screening) alongside study drug (lenabasum), and approximately half of patients received MMF (35). Though this inclusive design facilitated timely enrolment (37, 38), treatment with MMF was associated with greater improvement in ACR-CRISS score, change in mRSS, and change in ppFVC, compared with no MMF background immunosuppressant therapy (35).

The potential confounding effect of background MMF on treatment response could be mitigated by the requirement in the DAISY trial for MMF treatment to have been initiated ≥ 6 months prior to randomisation and at a stable dosage ≤ 3 g/day for ≥ 3 months prior to randomisation, in addition to including MMF use as a stratification factor for randomisation. While it could be considered a risk of the study design, including MMF as a background therapy will allow DAISY to provide meaningful information to treating clinicians managing SSc (e.g. rheumatologists, dermatologists, and pulmonologists) who wish to understand the role of anifrolumab alongside frequently prescribed standard therapies.

To conclude, DAISY is the first phase 3 trial to investigate a first-in-class IF-NAR1 antibody in patients with both diffuse and limited cutaneous SSc. The study design was informed by patients and utilises a composite endpoint that incorporates assessments of how patients feel and function along with clinical disease activity assessments. As it includes a broad patient population, permits the use of standard therapy, and spans a 2-year on-drug period, DAISY will be able to provide important insights into the potential of anifrolumab for SSc treatment.

Acknowledgements

This study was sponsored by Astra-Zeneca. Writing assistance by Katey Glunt, PhD, of JK Associates Inc., part of Avalere Health.

Competing interests

D. Khanna has received grant support from Amgen, BMS, and Boehringer Ingelheim to his institution; consulting fees from Amgen, AstraZeneca, Argenx, BMS, GSK, and Merck; personal fees from Janssen; meeting support from Boehringer Ingelheim; and participation on a Data Safety Monitoring or Advisory Board for AbbVie.

C.P. Denton has received grant support from AbbVie, Arxx, GSK, and Horizon; consulting fees from Acceleron, Arxx, AstraZeneca, Bayer, Boehringer Ingelheim, Certa, Corbus, CSL Behring, Galapagos, GSK, Horizon, Janssen, Lilly, Novartis, Roche, and Sanofi-Aventis.

S. Assassi has received grant support from aTyr, Boehringer Ingelheim, Janssen, and Scleroderma Research Foundation; personal fees from AstraZeneca, aTyr, Boehringer Ingelheim, CSL Behring, Merck, and TeneoFour.

M. Kuwana has received grant support from Boehringer Ingelheim and MBL; royalties or licenses from MBL; consulting fees from AstraZeneca, Boehringer Ingelheim, Chugai, and GSK; personal fees from Asahi Kasei Pharma, Boehringer Ingelheim, Chugai, and Janssen; and participation on a Data Safety Monitoring or Advisory Board for Argenx.

Y. Allanore has received consulting fees from AbbVie, AstraZeneca, Boehringer Ingelheim, Medsenic, Prometheus, and Topadur; support for attending meetings/travel from AstraZeneca and Boehringer Ingelheim.

R.T. Domsic has received consulting fees from Aisa Pharmaceuticals and AstraZeneca; and participation on a Data Safety Monitoring or Advisory Board for CSL Behring.

C. Kleoudis is employee of AstraZeneca. J. Xu, E. Csomor, C. Seo, M. Albulescu, R. Tummala and R.N. Kalyani are employees and shareholders of Astra-Zeneca.

H. Al-Mossawi has role of leadership or fiduciary role at Immunocore; a formerly a full-time employee of AstraZeneca, and holds stock or stock options from AstraZeneca and Immunocore.

F. Del Galdo has received grant support from AbbVie, AstraZeneca, Boehringer Ingelheim, Chemomab, Janssen, Mitsubishi-Tanabe, and NIHR; consulting fees from AbbVie, Arxx, Argenx, Astra-Zeneca, Boehringer-Ingelheim, GSK, Janssen, and Mitsubishi-Tanabe; personal fees from Boehringer Ingelheim, GSK, and Janssen.

References

- 1. DENTON CP, KHANNA D: Systemic sclerosis. Lancet 2017; 390(10103): 1685-99. https:// doi.org/10.1016/s0140-6736(17)30933-9
- 2. IACCARINO L, GATTO M, BETTIO S *et al.*: Overlap connective tissue disease syndromes. *Autoimmun Rev* 2013; 12(3): 363-73. https://doi.org/10.1016/j.autrev.2012.06.004
- KAKKAR V, ASSASSI S, ALLANORE Y et al.: Type 1 interferon activation in systemic sclerosis: a biomarker, a target or the culprit. Curr Opin Rheumatol 2022; 34(6): 357-64. https:// doi.org/10.1097/bor.00000000000000907
- 4. HIGGS BW, LIU Z, WHITE B et al.: Patients with systemic lupus erythematosus, myositis, rheumatoid arthritis and scleroderma share activation of a common type I interferon pathway. Ann Rheum Dis 2011; 70(11): 2029-36. https://doi.org/10.1136/ard.2011.150326
- SKAUG B, ASSASSI S: Type I interferon dysregulation in systemic sclerosis. *Cytokine* 2020; 132: 154635.

https://doi.org/10.1016/j.cyto.2018.12.018

- ASSASSI S, MAYES MD, ARNETT FC et al.: Systemic sclerosis and lupus: points in an interferon-mediated continuum. Arthritis Rheum 2010; 62(2): 589-98. https://doi.org/10.1002/art.27224
- LIU X, MAYES MD, TAN FK *et al.*: Correlation of interferon-inducible chemokine plasma levels with disease severity in systemic sclerosis. *Arthritis Rheum* 2013; 65(1): 226-35. https://doi.org/10.1002/art.37742
- ASSASSI S, WANG X, CHEN G et al.: Myeloablation followed by autologous stem cell transplantation normalises systemic sclerosis molecular signatures. Ann Rheum Dis 2019; 78(10): 1371-8. https://
- doi.org/10.1136/annrheumdis-2019-215770
 9. FURST D, KHANNA D, MATUCCI-CERINIC M et al.: Systemic sclerosis - continuing progress in developing clinical measures of response. J Rheumatol 2007; 34(5): 1194-200.
- BLACK CM, SILMAN AJ, HERRICK AI *et al.*: Interferon-alpha does not improve outcome at one year in patients with diffuse cutaneous scleroderma: results of a randomized, double-blind, placebo-controlled trial. *Arthritis Rheum* 1999; 42(2): 299-305. https://doi.org/ 10.1002/1529-0131(199902)42:2<299::aidanr12>3.0.co;2-r
- POSTAL M, VIVALDO JF, FERNANDEZ-RUIZ R, PAREDES JL, APPENZELLER S, NIEWOLD TB: Type I interferon in the pathogenesis of systemic lupus erythematosus. *Curr Opin Immunol* 2020; 67: 87-94. https://doi.org/10.1016/j.coi.2020.10.014
- FURIE R, KHAMASHTA M, MERRILL JT et al.: Anifrolumab, an anti-interferon-α receptor monoclonal antibody, in moderate-to-severe systemic lupus erythematosus. Arthritis Rheumatol 2017; 69(2): 376-86. https://doi.org/10.1002/art.39962
- RIGGS JM, HANNA RN, RAJAN B et al.: Characterisation of anifrolumab, a fully human anti-interferon receptor antagonist antibody for the treatment of systemic lupus erythematosus. Lupus Sci Med 2018; 5(1): e000261. https://doi.org/10.1136/lupus-2018-000261
- 14. ASTRAZENECA: SAPHNELO (anifrolumabfnia) [prescribing information]. AstraZeneca

Pharmaceuticals LP; 2021.

- 15. EMA: Saphnelo (anifrolumab). European Medicines Agency, 2024.
- 16. Saphnelo approved in Japan for systemic lupus erythematosus [press release]. 2021.
- ASTRAZENECA. Australian Product Information: Saphnelo. AstraZeneca Pharmaceuticals LP; 2022.
- MORAND EF, FURIE R, TANAKA Y *et al.*: Trial of anifrolumab in active systemic lupus erythematosus. *N Engl J Med* 2020; 382(3): 211-21.
 - https://doi.org/10.1056/nejmoa1912196
- FURIE RA, MORAND EF, BRUCE IN et al.: Type I interferon inhibitor anifrolumab in active systemic lupus erythematosus (TULIP-1): a randomised, controlled, phase 3 trial. Lancet Rheumatol 2019; 1(4): e208-e19. https:// doi.org/10.1016/S2665-9913(19)30076-1
- 20. KALUNIAN KC, FURIE R, MORAND EF et al.: A randomized, placebo-controlled Phase III extension trial of the long-term safety and tolerability of anifrolumab in active systemic lupus erythematosus. Arthritis Rheumatol 2023; 75(2): 253-65. https://doi.org/10.1002/art.42392
- 21. GOLDBERG A, GEPPERT T, SCHIOPU E et al.: Dose-escalation of human anti-interferon-alpha receptor monoclonal antibody MEDI-546 in subjects with systemic sclerosis: a phase 1, multicenter, open label study. Arthritis Res Ther 2014; 16(1): R57. https://doi.org/10.1186/ar4492
- 22. GUO X, HIGGS BW, BAY-JENSEN AC et al.: Suppression of T cell activation and collagen accumulation by an anti-IFNAR1 mAb, anifrolumab, in adult patients with systemic sclerosis. J Invest Dermatol 2015; 135(10): 2402-9. https://doi.org/10.1038/jid.2015.188
- 23. VAN DEN HOOGEN F, KHANNA D, FRANSEN J et al.: 2013 classification criteria for systemic sclerosis: An American College of Rheumatology/European League Against Rheumatism collaborative initiative. Arthritis Rheum 2013; 65(11): 2737-47. https://doi.org/10.1002/art.38098
- 24. LEROY EC, MEDSGER TA, JR.: Criteria for the classification of early systemic sclerosis. *J Rheumatol* 2001; 28(7): 1573-6.
- 25. KHANNA D, HUANG S, LIN CJF, SPINO C: New composite endpoint in early diffuse cutaneous systemic sclerosis: revisiting the provisional American College of Rheumatology Composite Response Index in Systemic Sclerosis. Ann Rheum Dis 2021; 80(5): 641-50. https://
- doi.org/10.1136/annrheumdis-2020-219100
 26. KHANNA D, BERROCAL VJ, GIANNINI EH et al.: The American College of Rheumatology provisional composite response index for clinical trials in early diffuse cutaneous systemic sclerosis. Arthritis Rheumatol 2016; 68(2): 299-311. https://doi.org/10.1002/art.39501
- 27. YAO Y, HIGGS BW, MOREHOUSE C et al.: Development of potential pharmacodynamic and diagnostic markers for anti-IFN-α monoclonal antibody trials in systemic lupus erythematosus. *Hum Genomics Proteomics* 2009; 2009. https://doi.org/10.4061/2009/374312
- 28. POPE JE, DENTON CP, JOHNSON SR, FER-NANDEZ-CODINA A, HUDSON M, NEVSKAYA

T: State-of-the-art evidence in the treatment of systemic sclerosis. *Nat Rev Rheumatol* 2023; 19(4): 212-26.

- https://doi.org/10.1038/s41584-023-00909-5 29. CHRISTMANN RB, SAMPAIO-BARROS P, STI-FANO G et al.: Association of Interferon- and transforming growth factor beta-regulated genes and macrophage activation with systemic sclerosis-related progressive lung fibrosis. Arthritis Rheumatol 2014; 66(3): 714-25. https://doi.org/10.1002/art.38288
- 30. FARINA G, LAFYATIS D, LEMAIRE R, LAFYA-TIS R: A four-gene biomarker predicts skin disease in patients with diffuse cutaneous systemic sclerosis. *Arthritis Rheum* 2010; 62(2): 580-8. https://doi.org/10.1002/art.27220
- 31. BRKIC Z, VAN BON L, COSSU M et al.: The interferon type I signature is present in systemic sclerosis before overt fibrosis and might contribute to its pathogenesis through high BAFF gene expression and high collagen synthesis. Ann Rheum Dis 2016; 75(8): 1567-73.
- 32. FRANTZ C, HUSCHER D, AVOUAC J et al.: Outcomes of limited cutaneous systemic sclerosis patients: Results on more than 12,000 patients from the EUSTAR database. Autoimmun Rev 2020; 19(2): 102452.
- https://doi.org/10.1016/j.autrev.2019.102452 33. ALLANORE Y: Limited cutaneous systemic sclerosis: the unfairly neglected subset. J Scleroderma Relat Disord 2016; 1(3): 241-6. https://doi.org/10.5301/jsrd.5000216
- 34. TASHKIN DP, ROTH MD, CLEMENTS PJ et al.: Mycophenolate mofetil versus oral cyclophosphamide in scleroderma-related interstitial lung disease (SLS II): a randomised controlled, double-blind, parallel group trial. Lancet Respir Med 2016; 4(9): 708-19. https://doi.org/10.1016/S2213-2600(16)30152-7
- 35. SPIERA R, KUWANA M, KHANNA D et al.: Efficacy and safety of lenabasum, a cannabinoid type 2 receptor agonist, in a phase 3 randomized trial in diffuse cutaneous systemic sclerosis. Arthritis Rheumatol 2023; 75(9): 1608-18. https://doi.org/10.1002/art.42510
- 36. CLEMENTS PJ, TASHKIN DP, ROTH MD et al.: The Scleroderma Lung Study II (SLS II) shows that both oral cyclophosphamide (CYC) and mycophenolate mofetil (MMF) are efficacious in treating progressive intersitial lung disease (ILD) in patients with systemic sclerosis (SSc). Arthritis Rheumatol 2015; 67 (Suppl 10).
- 37. DENTON C, BLOOM B, DGETLUCK N, WHITE B, SPIERA R: Baseline European patient demographics and disease characteristics in a phase 3 study of safety and efficacy of lenabasum, a CB2 agonist, in diffuse cutaneous systemic sclerosis. Ann Rheum Dis 2020; 79: 1580-1.
- 38. SPIERA R, KHANNA D, KUWANA M et al.: A randomised, double-blind, placebo-controlled phase 3 study of lenabasum in diffuse cutaneous systemic sclerosis: RESOLVE-1 design and rationale. *Clin Exp Rheumatol* 2021; 39(4): S124-33. https://
- doi.org/10.55563/clinexprheumatol/i80zh7
 39. KHANNA D, DENTON C, ASSASSI S *et al.*: P.336: A randomized, parallel-group, doubleblind, placebo-controlled phase 3 study of the type 1 interferon receptor antibody, anifrolumab, in systemic sclerosis: design and

rational of the DAISY trial. J Scleroderma Relat Disord 2024; 9 (Supplement): 61-395.

40. GRAHAM BL, STEENBRUGGEN I, MILLER MR *et al.*: Standardization of spirometry 2019 update: an official American Thoracic Society and European Respiratory Society technical statement. *Am J Respir Crit Care Med* 2019; 200(8): e70-e88.

https://doi.org/10.1164/rccm.201908-1590st

- 41. CLEMENTS PJ, LACHENBRUCH PA, SEIBOLD JR et al.: Skin thickness score in systemic sclerosis: an assessment of interobserver variability in 3 independent studies. J Rheumatol 1993; 20(11): 1892-6.
- 42. RAMEY DR, FRIES JF, SINGH G: Quality of life and pharmacoeconomics in clinical trials. *In* SPILKER B (Ed.): The Health Assessment Questionnaire 1995 -- Status and

Review. 2nd ed. Philadelphia, PA, USA, Lippincott-Raven, 1996, 227-37.

43. MAN A, CORREA JK, ZIEMEK J, SIMMS RW, FELSON DT, LAFYATIS R: Development and validation of a patient-reported outcome instrument for skin involvement in patients with systemic sclerosis. Ann Rheum Dis 2017; 76(8): 1374-80. https:// doi.org/10.1136/annrheumdis-2016-210534